Author’s response to reviews

Title: Complete Atrioventricular Block due to Timolol Eye Drops: A Case Report and Literature Review

Authors:

Zhuoying Wang (ingrid868@gmail.com)
Ian Denys (idenys@lsuhsc.edu)
Feng Chen (chenfengmail@yeah.net)
Lijie Cai (2484427377@qq.com)
Xuecui Wang (982016875@qq.com)
Daniel Kapusta (dkapus@lsuhsc.edu)
Yongliang Lv (lylv@sohu.com)
Juan Gao (jgao1@lsuhsc.edu)

Version: 4 Date: 29 Oct 2019

Author’s response to reviews:

Answer to Editor’s Comments

Please could you clarify if written informed consent was obtained from the patient for publication of this case report and any accompanying information, and if a copy of the consent form is available for review by the Editor of the journal. Please see our policies through the link below for more information:

https://www.biomedcentral.com/getpublished/editorial-policies#consent+for+publication

Response: When we were writing this case report, we obtained the informed consent for publication from the patient in 2018. This form is a standard form used by our hospital, but it is in Chinese. After we received the email from editor on 10/25/2019, the first author of this case report, Zhuoying Wang visited the patient on 10/28/2019, and explained the manuscript and informed consent for publication using the template provided by BMC Pharmacology and Toxicology to him. The patient was excited that his case will be accepted for publication and glad to give his permission again. Therefore, we have both the signed informed consent in BMC Pharmacology and Toxicology template in English and the original hospital standard informed consent in Chinese in the Supplemental Materials.